Author's response to reviews

Title: Can oral corticosteroids reduce the severity or duration of an acute cough, and the associated NHS and societal costs, in adults presenting to primary care? Study protocol of a randomised controlled trial

Authors:

Harriet E Downing (harriet.downing@bristol.ac.uk)
Fran Carroll (fran.carroll@bristol.ac.uk)
Sara T Brookes (s.t.brookes@bristol.ac.uk)
Sandra P Hollinghurst (s.p.hollinghurst@bristol.ac.uk)
David Timmins (david.timmins@phc.ox.ac.uk)
Elizabeth Orton (elizabeth.orton@nottingham.ac.uk)
Paul Little (p.little@soton.ac.uk)
Mike Moore (mvm198@soton.ac.uk)
Denise Kendrick (denise.kendrick@nottingham.ac.uk)
Anthony Harnden (anthony.harnden@phc.ox.ac.uk)
Kay Wang (kay.wang@phc.ox.ac.uk)
Matthew Thompson (matthew.thompson@phc.ox.ac.uk)
Margaret T May (m.t.may@bristol.ac.uk)
Alastair D Hay (alastair.hay@bristol.ac.uk)

Version: 2 Date: 31 December 2014

Author's response to reviews:

The revisions to the manuscript that have been made in response to the reviewer comments are described in the covering letter. The editorial comments have been addressed in the revised manuscript.